Author's response to reviews

Title: Mental health and wellbeing in spouses of persons with dementia: the Nord-Trondelag health study

Authors:

Helga Ask (heas@fhi.no)
Ellen M Langballe (ella@fhi.no)
Jostein Holmen (jostein.holmen@ntnu.no)
Geir Selbæk (geir.selbaek@aldringoghelse.no)
Ingvild Saltvedt (ingvild.saltvedt@ntnu.no)
Kristian Tambs (kpta@fhi.no)

Version: 2 Date: 22 July 2013

Author’s response to reviews:

a) An ethics statement is included in the Method section, stating that the project was approved by the Regional Ethichs Committee and that the research carried out is in compliance with the Helsinki Declaration.